Author's response to reviews

Title: Study protocol for improving asthma outcomes through cross-cultural communication training for physicians: a randomized trial of physician training

Authors:

Minal R Patel (minalrp@umich.edu)
Lara J Thomas (ljthomas@umich.edu)
Kausar Hafeez (khafeez@umich.edu)
Matthew Shankin (shankinm@umich.edu)
Margaret Wilkin (mwilkin@umich.edu)
Randall W Brown (rbrownmd@umich.edu)

Version: 3
Date: 22 May 2014

Author's response to reviews: see over
May 22, 2014

Natalie Pafitis
Executive Editor, *BMC Public Health*
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Dear Dr. Pafitis:

Thank you and the manuscript reviewers for the first round of feedback on our manuscript entitled “Study protocol for improving asthma outcomes through cross-cultural communication training for physicians: a randomized trial of physician training.” We appreciate the comments regarding their interest in the manuscript. We have revised the manuscript to include the date of the clinical trial registration at the end of the abstract. The revision now reads as follows: ClinicalTrials.gov: NCT01251523 December 1, 2010

I hope you find this paper suitable for the journal. Please contact me with questions or if further information is needed.

Sincerely,

Minal R. Patel, Ph.D, M.P.H
Assistant Professor
Department of Health Behavior and Health Education
University of Michigan School of Public Health